Author’s response to reviews

Title: AJCC 8th Edition Prognostic Staging provides no better discriminatory ability in prognosis than Anatomical Staging in triple negative breast cancer

Authors:

Jiehua He (hejh@sysucc.org.cn)
Julia Tsang (jystsang@cuhk.edu.hk)
Xiaodan Xu (xiaodanchelyn@foxmail.com)
Jibin Li (lijib@sysucc.org.cn)
Mei Li (limei@sysucc.org.cn)
Xue Chao (chaoxue@sysucc.org.cn)
Yuanyuan Xu (xuuy37@mail2.sysu.edu.cn)
Rongzhen Luo (luorzh@sysucc.org.cn)
Gary Tse (gmktse@gmail.com)
Peng Sun (sunpeng1@sysucc.org.cn)

Version: 2 Date: 21 Dec 2019

Author’s response to reviews:

Editor comments:

C1. Please only include one corresponding author on the title page. Please be sure to make the corresponding author the same as in the submission data located in the editorial manager. In this case, author Peng Sun is indicated as the corresponding author.
A1. Please see the changes on the title page in line 8 and line 24 of the revised version of the manuscript.

C2. Please add a Conclusion section heading to your manuscript.
A2. The conclusion section has been added in line 278-282 of the revised version of the manuscript. "Conclusion Our findings demonstrated that prognostic staging system did not provide better discriminatory ability in predicting TNBCs prognosis than anatomic staging system. The prognosis staging system should be further modified following the development of novel clinical therapies in TNBCs."

C3. Under your Ethics approval and consent to participate heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
A3. In the revised version of the manuscript, we stated that "The formal written informed consent was obtained from all individual participants included in the study." in line 299-300 under the ethics approval and consent to participate section.

C4. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

A4. The uploaded file do not include any any tracked changes, colored or highlighted text.

Many thanks for the kindly suggestions.